Author’s response to reviews

Title: Chronic administration of parecoxib exerts anxiolytic-like and memory enhancing effects and modulates synaptophysin expression in mice

Authors:

Bo Wang (277433307@qq.com)
Xin Jin (812823649@qq.com)
Xin Kuang (kxnhfy@126.com)
Shaowen Tian (tsw.neuro@126.com)

Version: 2 Date: 02 Nov 2017

Author’s response to reviews:

Dear Editor,

Thank you for your letter concerning our manuscript entitled Chronic administration of parecoxib exerts anxiolytic-like and memory enhancing effects and modulates synaptophysin expression in mice. We have thoroughly read all remarks by the Editor. Detailed answers to all specific remarks are provided herein.

1. As suggested by the Editor, the list of abbreviations was provided in our revised manuscript.

2. As suggested by the Editor, we declared that the datasets generated and analysed during the current study are available from the corresponding author on reasonable request in the Availability of Data and Materials section.

3. As suggested by the Editor, we declared that the funding sources had no further role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript in the Funding section.

4. As suggested by the Editor, a clean version of the manuscript was resubmitted.

Enclosed please find the revised manuscript. We hope that the manuscript will eventually be published in BMC Anesthesiology.

Sincerely yours,

Xin Kuang
Address: Department of Anesthesiology, First Affiliated Hospital, University of South China, Hengyang, Hunan, 421001, P.R China